Author’s response to reviews

Title: Mistaken perception of lipid intake and its effects: a randomized trial

Authors:

Mariana Menezes (marysnut@gmail.com)
Sueli Mingoti (suelimngt@gmail.com)
Raquel Mendonça (raqueldmendonca@gmail.com)
Aline Lopes (alinelopesenf@gmail.com)

Version: 6 Date: 06 Jul 2017

Author’s response to reviews:

1) Please send a "clean" version of the manuscript meaning that all text is black.

Done. We have put all manuscript in black. The new version is named “Manuscript after review – clean version 06-17”.

2) Please send the revised manuscript ONLY when you have a Trial Registration Number in the manuscript which should appear at the end of the abstract with the date of registration.

We are sending the supposed ready manuscript, as we did not sent it last time (because we were asked to “only submit the revised version when the trial registration is received”), but the Journal asked about the submission’s delay.

We would like to highlight that the process with the registration is almost done and we believe we will have the registration number in the next days. We will inform the journal as soon as we receive it.

3) Please correct the typo for the study protocol approval date on page 18 line 422.

We didn’t mean to state the date of study protocol approval, but the study protocol approval reference number, as guided by the Journal: “Ethics approval and consent to participate: Manuscripts reporting studies involving human participants, human data or human tissue must include the name of the ethics committee that approved the study and the committee’s reference number if appropriate.”

We thank the editor and reviewers for the effort they have put into this report.